Author’s response to reviews

Title: Common variable immunodeficiency syndrome with right aortic arch; a case report

Authors:

Dr Didem A Yalcin (adidyal@yahoo.com)
Nese Aydemir (naydemir@pamukkale.edu.tr)
Huseyin Turgut (hturgut@pamukkale.edu.tr)
Dr Hakan R Erbay (rherbay@pamukkale.edu.tr)
Nevzat A Yalcin (anyalcin@akdeniz.edu.tr)

Version: 6 Date: 28 Jan 2004

I changed the block capitals in the section headings in the abstract to sentence case.

I removed the copy of the consent forms from the additional files.

I will fax the consent forms in the soonest possible time.

Thanks for your attention.

Dr.A.D.Yalcin